Check for updates

OPEN ACCESS

EDITED BY Hêriş Golpîra, Islamic Azad University, Iran

REVIEWED BY

Ángel Acevedo-Duque, Autonomous University of Chile, Chile Ming Lang Tseng, Asia University, Taiwan Baogui Xin, Shandong University of Science and Technology, China

*CORRESPONDENCE Guocheng Zhang, ☑ zhchen1126@gmail.com

RECEIVED 22 May 2023 ACCEPTED 23 October 2023 PUBLISHED 02 November 2023

CITATION

Wang M and Zhang G (2023), What motivates firms to adopt a green supply chain and how much does it matter? *Front. Environ. Sci.* 11:1227008. doi: 10.3389/fenvs.2023.1227008

COPYRIGHT

© 2023 Wang and Zhang. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

What motivates firms to adopt a green supply chain and how much does it matter?

Mengmeng Wang and Guocheng Zhang*

College of Business, Gachon University, Seongnam, Republic of Korea

This study integrates the elaboration likelihood model, institutional theory, and the image management perspective to identify the internal and external forces that can effectively drive firms to adopt a green supply chain. Using survey data from a sample of 246 firms in the Chinese manufacturing sector, we empirically examine how the internal green process and green product innovation of firms and different types of external institutional pressures contribute to their adoption of a green supply chain, which in turn can explain the variation in their environmental and new-product development performance. Results of our structural equation modeling analysis reveal that the internal green process and green product innovation of firms and three types of external institutional pressures (i.e., coercive, normative, and mimetic pressures) play an important role in driving green supply chain adoption. Moreover, firms can improve their environmental and new-product development performance by adopting a green supply chain. In addition, though the image management motivation of firms tends to play a negative moderating role in the relationship between their green process innovation and green supply chain adoption, such motivation positively moderates the contribution of coercive pressure to their green supply chain adoption. Our findings can help explain why emerging-market firms tend to adopt a green supply chain and do not follow the pattern predicted by classical strategic management theories. The results of our study clearly demonstrate the determinants and consequences of the adoption of a green supply chain of firms and confirm that their green supply chain adoption can exert a positive effect on their environmental and new-product development performance. Overall, our research highlights the importance of recognizing internal innovation and external institutional forces in driving the green supply chain adoption of firms, which in turn will contribute to their environmental and new-product development performance. In this regard, our study extends the stream of research on green supply chains and sheds new light on the importance of abandoning the conventional supply chain and adopting a green one. Our study also provides important implications for research, practice, and policymaking.

KEYWORDS

green supply chain, green process innovation, green product innovation, institutional pressure, environmental performance, new-product development performance

1 Introduction

With the growing global population and continuous economic development, problems related to natural resources and the environment have become increasingly serious. Resource shortage, environmental pollution, and climate change have become the main bottlenecks restricting the sustainable development of the global economy. Accordingly, the

establishment of a collaborative governance system for reducing pollution and carbon emissions has become a global objective, to which major countries have responded by actively promoting the creation and improvement of green supply chain systems (Green et al., 2012; Khan et al., 2023). At present, the traditional supply chain has yet to completely detach itself from the high-input, highconsumption, and high-emissions development mode; thus, ecological and environmental problems have become increasingly prominent, and the green development situation remains serious. By contrast, a green supply chain can integrate environmental protection ideas with green procurement, green manufacturing, green distribution, reverse logistics, and other links and covers the processes of upstream procurement, production, downstream logistics, and recycling, thereby minimizing the production of hazardous chemicals, emissions, solid wastes, and other toxic substances in the whole supply chain (Chin et al., 2015; Micheli et al., 2020).

The literature presents different perspectives to explain why firms should establish a green supply chain. For example, using data from manufacturing firms, Cheng and Sheu (2012) examined the effect of relationship orientation on interorganizational strategy quality in green supply chains and demonstrated the important role of opportunistic behavior in negatively moderating such a relationship. Most important, the possible opportunistic behavior of some players in the green supply chain management may lead to wastage of various resources and thus reduce the possible benefits of the supply chain integration (Wong et al., 2021). Moreover, the use of appropriate governance mechanisms may reduce possible opportunistic behavior and enhance commitment and trust among the players within a supply chain, which may lead to high-quality green supply chains (Li et al., 2014; Yang et al., 2021). In addition, a couple of studies attempted to extend this stream of research by examining the importance of designing a closed-loop supply chain and implementing circular economy strategies (Amir et al., 2022; Tavana et al., 2022) or spread green ideologies across supply chain practices to help firms gain a competitive advantage (e.g., Zaid et al., 2018; Wang et al., 2020). However, despite the interest and efforts in exploring the mechanisms and importance of the development and enhancement of green supply chains and the considerable progress, prior research on green supply chains lacks a systematic conceptual framework that can extend the field's reach to the current dynamic environment. Given the potential benefits of the facilitation of green process and product innovation to meet customers' needs, some scholars argued that firms can establish a green supply chain through such innovation (Silva et al., 2019). Green process innovation emphasizes a firm's ability to improve its existing processes and develop new modes to achieve energy conservation, emissions reduction, pollution prevention, and enhanced energy and resource utilization efficiency in its entire production process (Huang and Li, 2015; Tian and Wang, 2019). This type of innovation can improve firms' environmental and corporate financial performance (Xie et al., 2019; Wang and Liu, 2022). Meanwhile, green product innovation can enable firms to meet customers' environmental protection demands, focus on the use and recycling of environment-friendly materials, reduce raw material and energy wastage in compliance with environmental regulations, and prevent health and safety risks (Chen et al., 2017).

Firms can adopt green designs, choose green materials, implement green procurement processes, promote green packaging, and advocate green transportation and waste product recycling to realize green environmental protection in their product cycle and minimize the negative impact of their products on the environment (Tariq et al., 2017). In addition, the government, customers, competitors, and other stakeholders can exert coercive, normative, and mimetic pressures on firms to promote green supply chain practices (Zhu and Sarkis, 2007). Governments have implemented environmental legislation, cap-and-trade laws, lowcarbon subsidies, and other measures at the macro level to guide firms in their adoption of a green supply chain (Mondal and Giri, 2022a; b). The establishment of a green supply chain is inevitable, but some challenges and opportunities must be considered, because a supply chain involves multiple links and interests. If firms follow national carbon emissions reduction standards, then emissions reduction costs will increase, and the final consumers will bear the high retail and wholesale prices (Cheng et al., 2017). Moreover, firms' emissions reduction is closely related to its cost. Therefore, the high cost of green technologies can hinder most firms from adopting a green supply chain to reduce their emissions, because it may substantially increase their costs (Govindan et al., 2014; Waltho et al., 2019). In addition to the risk of design defects in their green process (Rostamzadeh et al., 2018), failure in their machines or facilities (Rostamzadeh et al., 2018), and uncertainties in their supply of key green raw materials, firms can encounter other operational risks, supply risks, product recovery risks, financial risks, demand risks, and government- and organization-related risks in the process of implementing a green supply chain (Pourjavad and Shahin, 2020). In the face of such risks, firms may be unable to manage a green supply chain and operate with green supply chain resilience; thus, their suppliers, distributors, and logistics service providers may lack coordination (Brusset and Teller, 2017).

In summary, most previous studies that explored how to promote the green upgrading of supply chains tended to focus largely on external factors, such as external drivers or the resistance encountered by firms, in the process of adopting a green supply chain, and those to effectively understand the process of the adoption of green supply chain practices from the internal and external perspectives of such firms have yet to be conducted. Despite the importance of adopting a green supply chain to achieve sustainable development, firms are hesitant to abandon their traditional supply chain in favor of a green one owing to the risks and obstacles involved. In this study, to fill the research gaps in the literature and boost the confidence of firms and persuade other firms to completely abandon their traditional supply chain in favor of a green one, we address three interrelated research questions. 1) What are the internal and external forces that can drive firms to adopt green supply chain practices? 2) How can the adoption of green supply chain practices help firms improve their environmental and new-product development performance? 3) Under what conditions can internal and external forces influence a firm's adoption of a green supply chain? Building on the elaboration likelihood model (ELM) and image management and institutional perspectives, we propose a comprehensive model. By considering the green product and green process innovation proactively adopted by firms as the

central path and the coercive, normative, and mimetic pressures imposed by the government, customers, and competitors as the peripheral path, we aim to investigate the most effective persuasion methods for firms to change and adopt a green supply chain from internal and external perspectives and empirically analyze whether the change and adoption of a green supply chain can enhance their environmental and new-product performance. Firms may be motivated by maximum rewards and minimal punishment to effectively manage their image; thus, we also incorporate firm image management motivation into the conceptual framework by examining its moderating role in shaping the effect of internal and external drivers on green supply chain adoption. We hope to enrich the literature by extending the scope of green supply chain research and providing important guidelines on how to persuade firms to abandon their conventional supply chain and actively adopt a green one.

The rest of this paper is organized as follows: Section 2 introduces the theoretical background and hypotheses, Section 3 presents an overview of the data and research methods, Section 4 reports the empirical findings, Section 5 focuses on the discussion and implications, and Section 6 concludes the study and provides suggestions for future research avenues.

2 Theoretical background and hypothesis development

As a persuasion theory that describes attitudinal change, the ELM posits that people change their attitude through the central and peripheral paths of persuasion. In the central path, the change in the attitude of an individual is processed through the thorough processing and careful review of the available information. Individuals who follow this path generally engage in active participation, which is regarded as the most influential persuasion paradigm. Meanwhile, individuals who follow the peripheral path are satisfied with their current cognition and unwilling to spend energy analyzing information and can easily be persuaded by peripheral factors. Therefore, such individuals generally engage in passive participation (Petty and Cacioppo, 1986; Crano and Prislin, 2006). The ELM has been widely cited in academic research for its ability to clearly illustrate how individuals can be persuaded to change their attitude (Pasadeos et al., 2008). The use of this model has also been expanded, from the consumer sector to other areas, including e-commerce platforms (Wang and Yang, 2021), start-up financing (Moradi and Badrinarayanan, 2021), and crowdfunding (Ba et al., 2022). Although the ELM has become ubiquitous in practice, the media environment and consumer choices have drastically changed since the advent of the Internet. Therefore, the ELM must be rerecognized and improved and not be taken as the sole theoretical basis for persuasion (J. Kitchen et al., 2014).

We also refer to image management theory in building our model. The theory illustrates the process through which people try to manage and control their image in the mind of others. Image management motivation refers to the degree to which people manipulate and control their image in the mind of others. The stronger the relation between individuals and their motivation, the more valuable and closer their motivation, and the stronger their image management motivation. Individuals consciously control their behavior to influence the impression of their target audience (Leary and Kowalski, 1990). Organizations exert considerable effort in creating a positive image to satisfy others and be positively viewed by their stakeholders (Perks et al., 2013) and adopt various methods, such as self-promotion and ingratiation (Bolino and Turnley, 1999), the rational use of social media (Schniederjans et al., 2013), and camouflage (Benson et al., 2015).

We also built our model on institutional theory, which emphasizes social factors that can influence organizational behavior, including coercive, normative, and mimetic pressures. Among the different types of pressures, coercive pressure refers mainly to formal or informal pressure exerted by the government to force firms to change their behavior, such as through rewards or punishment policies (Chu et al., 2018). Under mandatory policies, the development of firms with low resource utilization efficiency and high pollution discharge will be restricted, and those with substandard wastewater and waste gas discharge may be punished. The government can also exert external pressure on firms by providing tax reliefs, subsidies, low-interest-rate financing, and other incentives (Nie et al., 2016; Russi et al., 2016). Normative pressure stems from the standards set by industry associations, media and public supervision, and other sources, such as educational institutions, industry groups, nongovernment organizations, suppliers, and customers (Zhu et al., 2013; Chu et al., 2017). Meanwhile, mimetic pressure mainly comes from competitors. When companies are uncertain about their technologies, goals, or circumstances, they may imitate the behavior of other firms to solve similar problems (DiMaggio and Powell, 1983). Similarly, if a major competitor that adopted a green strategy is favored by customers, then other companies in the same industry will likely follow suit (Chu et al., 2018; Qin et al., 2021). Previous studies showed that the three types of pressures can positively influence e-commerce transformation intention (Lin et al., 2020) and encourage firms to adopt sustainable supply chain management practices (Dai et al., 2021). Specifically, the three types of pressures can play a positive role in improving firms' energy-saving ability (Zhang et al., 2022) and green supply chain management (Bag et al., 2022) and the circular economy (Arranz et al., 2022).

Our study explores the methods used to persuade firms to adopt a green supply chain, the results of their green supply chain adoption, and the situation that drove the firms to switch to a different supply chain. Our study initially refers to the ELM by taking the two measures of green product innovation and green process innovation, which firms positively adopt, as the central route. Meanwhile, by referring to institutional theory and image management theory, our study takes the coercive, normative, and mimetic pressures that firms passively bear as the peripheral route. Our study also uses image management motivation as a moderating variable to understand the causes and consequences of firms' green supply chain adoption. As shown in Figure 1, Hypotheses 1 and 2 indicate that green process innovation and green product innovation, as the central path, will positively influence firms' adoption of a green supply chain, respectively, and Hypotheses 3, 4, and 5 indicate that coercive, normative, and mimetic pressures, as the peripheral path, will positively influence firms' adoption of a green supply chain, respectively. Given the willingness of firms to manipulate and control their image in the mind of others, Hypotheses 6a and 6b propose that image management motivation will play a moderating role in the relationship between green process innovation and green product innovation



and the adoption of a green supply chain. Meanwhile, Hypotheses 6c, 6d, and 6e propose that image management motivation will play a moderating role in the effect of coercive, normative, and mimetic pressures on the adoption of a green supply chain, and Hypotheses 7 and 8 propose that firms' adoption of a green supply chain can positively influence their environmental and new-product development performance.

2.1 Green innovation and green supply chain adoption

Firms mainly conduct green supply chain management to solve the rocky relationship between their supply chain management and natural environment. Guided by sustainability, green elements have been integrated into supply chain management, including product design, material purchasing, the manufacturing process, the delivery of finished products, recycling, and other links. Green supply chain management is considered to be an effective method for reducing environmental hazards while meeting the ecological needs of end users (Srivastava, 2007; Bag et al., 2022). The construction of a green supply chain requires the joint efforts of all the firms in the supply chain. The green innovation practices actively conducted by firms mainly include green product innovation and green process innovation. By using nontoxic compounds or biodegradable materials, firms can improve their product design and recycling through green product innovation and thus reduce the negative impact of their products on the environment. Meanwhile, firms can introduce terminal and clean technologies into their product manufacturing process, change their existing production process, and reduce waste, waste gas, wastewater, and other pollutants generated in each stage of the production process through green process innovation to improve their resource and energy utilization efficiency (Severo et al., 2017; Xie et al., 2019).

As a positive environmental protection practice, if green product innovation integrates environmental protection factors into the product design and packaging, then green process innovation environmental protection factors into the integrates manufacturing process (Dangelico et al., 2016). Green product innovation and green process innovation can help firms not only gain a competitive advantage but also eliminate pollutant emissions from the source in the supply chain (Chen and Liu, 2020). Moreover, the implementation of green product and green process innovation practices may increase firms' likelihood of and effectiveness in adopting a green supply chain, which can provide them with new ideas (Chen et al., 2006). The green product and green process innovation practices of each firm in the supply chain are expected to prompt other firms to switch from the traditional to a green supply chain. Based on the above discussion, we propose the following hypotheses.

Hypothesis 1. Green process innovation will have a positive impact on a firm's green supply chain adoption.

Hypothesis 2. Green product innovation will have a positive impact on a firm's green supply chain adoption.

2.2 Institutional pressure and green supply chain adoption

According to institutional theory, firms' green product practices are generally affected by coercive, normative, and mimetic pressures (Huang and Chen, 2022). Coercive pressure is typically regarded as a form of governance pressure that forces companies to comply with various government regulations and policies (Chu et al., 2018). Under mandatory policies, the development of firms with low resource utilization efficiency and high pollution discharge will be restricted, and firms with substandard wastewater and waste gas discharge may be punished. The government can also exert external pressure on firms by providing tax reliefs, subsidies, low-interest-rate financing, and other incentives (Nie et al., 2016; Russi et al., 2016). Normative pressure stems from a wide range of sources, including educational institutions, industry groups, nongovernment organizations, suppliers, and customers (Zhu et al., 2013; Chu et al., 2017), whereas mimetic pressure mainly comes from competitors. If a major competitor that adopted a green strategy is favored by customers, then other companies in the same industry will likely follow suit (Chu et al., 2018; Qin et al., 2021).

The efforts of governments around the world in addressing environmental deterioration issues have increased the environmental protection awareness of industries and the popularity of green supply chain adoption. Therefore, as coercive, normative, and mimetic pressures increase, firms will likely adopt a green supply chain; thus, we propose the following hypotheses.

Hypothesis 3. Coercive pressure will have a positive impact on a firm's green supply chain adoption.

Hypothesis 4. Normative pressure will have a positive impact on a firm's green supply chain adoption.

Hypothesis 5. Mimetic pressure will have a positive impact on a firm's green supply chain adoption.

2.3 Interactive effect of green innovation and image management motivation

Image management motivation mainly includes motivation to improve one's social status, promote the development of an ideal identity, obtain valuable resources, protect oneself from potential harm, and avoid social exclusion (Leary and Kowalski, 1990). People with high image management motivation will pay little attention to themselves, focus considerably on the interests of others and their social environment, engage in prosocial behavior, and tend to practice green consumption (Zhang et al., 2019). At the organizational level, owing to locational and situational differences, firms will tend to adopt different image management strategies to influence how they are viewed by others. Under the influence of different image management motivations, firms will project themselves differently depending on the situation to enhance their corporate image (Bolino et al., 2008). To leave a deep and lasting impression on their stakeholders, firms will attach considerable importance to their market, employer, financial, and CSR images and spare no expense in investing in social capital, product development and diversification, public relations, and social response (Highhouse et al., 2009). In terms of product development, with the increasing environmental awareness of the public, firms' increased investment in green products and green processes may not only prevent the occurrence of potential crises, such as environmental protests and legal penalties, but also raise

consumers' expectations on their environment friendliness and sustainability and hence leave a favorable green impression on their stakeholders (Chen, 2010). Firms with a positive green image can gain substantial economic returns from their green product and green process innovation (Xie et al., 2019); thus, other firms may shift from the traditional to a green supply chain. The creation of a green image is one of the corporate image management motivations of firms. When firms have high green image management motivation, they may be willing to adopt a green supply chain through green product and green process innovation. Based on the above discussion, we propose the following hypotheses.

Hypothesis 6a. A firm's image management motivation will play a positive role in moderating the relationship between its green process innovation and green supply chain adoption.

Hypothesis 6b. A firm's image management motivation will play a positive role in moderating the relationship between its green product innovation and green supply chain adoption.

2.4 Interactive effect of institutional pressure and image management motivation

When making strategic decisions, firms will bear coercive, normative, and mimetic pressures from the government, institutions, their partners, and their suppliers. Whether such pressures can change firms' behavior will depend on the cognition of their manager (Liang et al., 2007; Ogbanufe et al., 2021). Firms' image management strategy will change with the pressure exerted by their stakeholders. With increasing pressure, firms may adopt four different image management strategies, that is, clarifying their position, clarifying their initial position, repairing their image, and adjusting their position. Firms will not easily fold under pressure but will gradually yield when the pressure reaches a certain level and then move from symbolic to substantive action (D. van Halderen et al., 2016). Although firms' environmental awareness has increased owing to regulatory, competitive, and marketing pressures, such awareness will not immediately lead to the adoption of green supply chain practices (Zhu et al., 2005). When firms' image management motivation increases further, they will be forced to forgo their self-concern and address the interests of others and their social environment and thus engage in green-oriented behaviors (Zhang et al., 2019). In other words, only when firms have high image management motivation will they be forced to adopt a green supply chain under coercive, normative, and mimetic pressures. Based on the above discussion, we propose the following hypotheses.

Hypothesis 6c. Image management motivation will positively moderate the relationship between coercive pressure and a firm's green supply chain adoption.

Hypothesis 6d. Image management motivation will positively moderate the relationship between normative pressure and a firm's green supply chain adoption.

Hypothesis 6e. Image management motivation will positively moderate the relationship between mimetic pressure and a firm's green supply chain adoption.

2.5 Green supply chain adoption and environmental performance

Environmental performance refers to the achievement and effect of environmental pollution management on the business activities of a firm, including the reduction of wastewater, waste gas, waste, and other harmful substances, to improve its material and energy use efficiency (Nakashima et al., 2006). Firms can build a green supply chain integration mechanism that incorporates environmental standards with the code of conduct of their employees, their business decisions, resource management, and other links. Through information sharing, collaboration, and other internal and external integration mechanisms, firms in a supply chain can improve their internal, supplier, and customer informationprocessing capabilities; reduce uncertainties in their green product and green process innovation; and indirectly improve their environmental performance (Wong et al., 2020). At the same time, green supply chain management practices, such as green procurement, investment recovery, and customer participation in environmental issues, can help improve firms' environmental performance (Zhu et al., 2012). In summary, when firms switch from the traditional to a green supply chain, they will be expected to show a marked improvement in their environmental performance. Thus, we propose the following hypothesis.

Hypothesis 7. A firm's green supply chain adoption will have a positive impact on its environmental performance.

2.6 Green supply chain adoption and newproduct development performance

The rapid changes in the market have substantially intensified competition among firms. Only by constantly developing new products can a firm meet the needs of and create value for its customers (Kavadias and Ulrich, 2020). New-product development performance refers to the extent to which a firm can successfully sell new products. This type of performance includes the financial and nonfinancial results of a firm attempting to develop new products and can be measured using multiple indicators, such as profit, return on investment, market share, and sales revenue (Najafi Tavani et al., 2013). Green supply chain management is viewed by many firms as a future development direction. To meet the environmental protection needs of their partners and customers, firms must change their new-product development strategies (Millson and Wilemon, 2006). In addition, firms should adopt green supply chain management practices in their green procurement, ecological design, internal environmental management, customer cooperation, and investment recovery to achieve a circular economy (Bag et al., 2022). When firms switch to green supply chain management, they will be expected to frequently introduce green products into the market. By introducing new green products that can meet the expectations and demands of the government, the society, and the public, firms can substantially improve their new-product development performance. Thus, we propose the following hypothesis.

Hypothesis 8. A firm's green supply chain adoption will positively affect its new-product development performance.

3 Methodology

3.1 Sampling and data collection

To examine the proposed hypotheses, we collected data on manufacturing firms in China through a survey. We believe that China is an ideal research setting to examine our conceptual framework. First, China is currently the second-largest economy in the world in terms of the nominal GDP and the largest emerging economy, accounting for a large proportion of the global economic growth. According to the World Bank, China is the largest contributor to world economic growth by contributing around 39%, on average, to the global economic growth during the period of 2013-2021, which exceeded the total contribution of the G7 economies. Second, China has become one of the most crucial and influential economies in the world in facilitating and leading green technology innovation. China implemented a series of policies to encourage and support market-oriented green technology innovation activities. The market-oriented green technology innovation system is expected to be largely improved by 2025 owing to China's green technology innovation for its green and low-carbon development. Last, as China aims to pursue lowcarbon and sustainable growth in the long term, many firms are embracing China's rapid transition to a green economy and thus investing considerably in green innovation to seize new business opportunities. Hence, an increasing number of firms in China are improving their green innovation capabilities as a response to the country's long-term goal of sustainable development.

To collect the data, we carefully designed the survey process. We randomly selected a sample of 600 firms from a list of Chinese manufacturing firms offered by a marketing research company. To develop the survey questionnaire, we first created an English questionnaire and then translated it into Chinese, with the assistance of two professional bilingual translators. To further ensure conceptual equivalence between the original English and translated Chinese versions of the questionnaire, we back translated the Chinese version into English, with the help of two additional bilingual translators (Brislin, 1980). In addition, before formally administering the survey, we conducted a series of in-depth interviews with senior managers of the Chinese firms to check the validity of our measures, then further modified some questionnaire items based on the feedback from the interviews. Conducting a survey on and collecting reliable data from firms in China are known to be challenging. The development of a positive relationship with the respondents is important to enhance the response rate and improve the quality of the survey data (Peng and Luo, 2000). Thus, we hired a highly renowned marketing research company with extensive experience in collecting firm data to conduct the formal survey and maintained a positive

relationship with the firms in the local Chinese market. With our careful survey procedure, we received a total of 358 responses. We obtained 246 useable responses after excluding 12 responses that had a number of missing values for some key variables, which represented an effective response rate of 41.0%.

Nonresponse bias may occur in a survey-based research; hence, we checked for potential nonresponse bias by comparing the early and late responding firms in terms of key firm characteristics (e.g., number of employees and firm age). The results of the t-statistics indicated that no statistically significant differences existed between the early and late responding firms in terms of the number of employees and firm age. Therefore, nonresponse bias was not likely a serious concern in our study (Armstrong and Overton, 1977). In addition, we checked for possible common method variance (CMV), which may emerge in the collection of self-reported data through a survey. Therefore, we performed Harman's one-factor test by conducting principal component factor analysis, with all the items for the multiple-item constructs included. The results of the unrotated factor analysis demonstrated that no general apparent factor emerged in the factor analysis and accounted for most of the variance (i.e., greater than 50%), with the first factor accounting for only 48.3% of the total variance in the data. Therefore, CMV was not likely a serious concern in our study.

3.2 Variables and measurement

To measure the variables, namely, the dependent, independent, and moderating variables, we used multiple-item seven-point Likert scales (1 = strongly disagree to 7 = strongly agree). All the measurement scales were well developed and widely used in the literature, which we further modified for our study.

Following prior research (e.g., Chen et al., 2006; Xie et al., 2019; Wang et al., 2021; Zameer et al., 2021), we measured green process innovation using three items asking the respondents to assess their level of green process innovation. Similar to prior studies (e.g., Chang, 2018; Xie et al., 2019; Chen and Liu, 2020; Wang et al., 2021), we used a fiveitem scale to measure the firms' degree of green product innovation. To measure the three types of pressures (i.e., coercive pressure, normative pressure, and mimetic pressure), we systematically reviewed relevant prior studies. Following the literature (e.g., Dubey et al., 2016; Bag et al., 2022), we adopted four items to measure the degree of perceived coercive pressure and normative pressure and three items to measure the degree of perceived mimetic pressure. Following prior studies (e.g., Sarkis et al., 2010; Mitra and Datta, 2014; Asif et al., 2020), we measured the firms' green supply chain adoption using seven items. To measure the degree of the firms' image management motivation, we adopted four items from the literature (e.g., Yun et al., 2007; Philp and Nepomuceno, 2020; Cheng et al., 2022). Furthermore, to measure the firm performance outcomes, we used two subdimensional performance measures (i.e., environmental performance and new-product development performance). Following prior studies (e.g., Seman et al., 2019; Zameer et al., 2021), we adopted five items to measure the firms' environmental performance. Following prior studies (e.g., Jeong et al., 2006; Xiao et al., 2021), we adopted six items to measure the firms' new-product development performance. The items asked the respondents to assess the broad dimensions of their firm's new-product development performance relative to that of their main competitors in the industry over the past 3 years. In addition, we

incorporated a number of controls into the analysis to rule out any alternative explanations, including firm size, firm age, and industry type. Following prior research (e.g., Park and Xiao, 2020; Xiao et al., 2021), we measured firm size using the logarithm of the number of employees and firm age using the number of years since the establishment of the firm. To control for the potential effect of industrial differences, we created a dummy variable that took the value of 1 if the firm domain was industrial and 0 for other domains (Park and Xiao, 2021).

4 Empirical analyses and results

4.1 Construct reliability and validity

We empirically tested the hypotheses by performing partial least squares structural equation modeling with SmartPLS (Richter et al., 2016). We first examined the reliability and validity of the measures before testing the hypotheses. The results are reported in Table 1, which presents the mean, standard deviation, factor loading, and reliability and validity of each construct and its indicators. As shown in Table 1, the factor loading of all the constructs was statistically significant and higher than 0.80, which indicated the strong reliability of the measurement model (Chin, 1998; Hulland, 1999). In addition, we checked the reliability of each construct by calculating its Cronbach's alpha and composite reliability (CR) values and found that all the Cronbach's alpha and CR values were higher than 0.80, which clearly exceeds the 0.70 cutoff (Nunnally, 1978; Fornell and Larcker, 1981) and provided further evidence of the reliability of the measurement model. To further check the reliability and convergent validity of the measures, we adopted the approach recommended by Fornell and Larcker (1981); that is, we calculated the average variance extracted (AVE) value of each construct. The results in Table 1 demonstrated that the AVE value of all the constructs was higher than 0.7, which exceeds the 0.50 cutoff and provided strong evidence of the adequate convergent validity and reliability of all the measures used in our study (Fornell and Larcker, 1981).

To assess the discriminant validity of the constructs used in the study, we compared the square root of the AVE of each construct and the absolute value of the correlation between a construct and all the other constructs. The results of the discriminant validity testing are reported in Table 2, which suggested that the square root of the AVE of each construct was greater than the absolute value of the correlation between a construct and all the other constructs, which provided strong evidence of the adequate discriminant validity of the measures we adopted for our study (Fornell and Larcker, 1981). Following the approach recommended by Henseler et al. (2015), we further checked the discriminant validity of the measures by assessing the heterotrait-monotrait ratio (HTMT) of the correlations. The results of our HTMT testing indicated that all the HTMT correlation values were not higher than 0.85, which provided further evidence of the adequate discriminant validity of all the measures we used in our study. Last, we assessed the predictive validity of each latent construct using Stone-Geisser's Q2 (Stone, 1974; Geisser, 1975). The results of our predictive validity assessment suggested that the cross-validated communality and redundancy values were greater than zero, which provided strong evidence of the predictive validity of the model (Fornell and Cha, 1994).

TABLE 1 Descriptive statistics and validity assessments.

	1		1
Construct and indicators	Mean	STD	FL
Green process innovation (AVE = 0.825, alpha = 0.894, CR = 0.934)			
Our firm is engaged in processes that reduce the emission of hazardous material	5.207	1.162	0.912
Our firm is engaged in processes that reduce consumption of electricity, water, gas, and petroleum	5.240	1.170	0.896
Our firm is engaged in processes that recycle, reuse, and reproduce material and decrease the use of raw material	5.386	1.197	0.918
Green product innovation (AVE = 0.799, alpha = 0.937, CR = 0.952)			
Our firm chooses the materials of the product that produce the least amount of pollution for conducting product development or design	5.419	1.331	0.894
Our firm chooses the materials of the product that can reduce the harmful effects to health	5.451	1.231	0.895
Our firm chooses the materials of the product that consume the least amount of energy and resources for conducting product development or design	5.329	1.307	0.913
Our firm uses the least amount of materials to comprise the product for conducting product development or design	5.329	1.207	0.888
Our firm would circumspectly deliberate whether the product is easy to recycle, reuse, and decompose for conducting product development or design	5.451	1.261	0.878
Coercive pressure (AVE = 0.826, alpha = 0.930, CR = 0.950)			
There are a large number of environmental regulations or restrictions imposed by the government on our firm's industry	5.337	1.357	0.908
There are frequent government inspections or audits on our firm to ensure that the firm is in compliance with environmental laws and regulations	5.362	1.378	0.894
Our firm tries to reduce or avoid the threat of current or future internal environmental legislations through adopting green innovation	5.289	1.260	0.908
Financial incentives offered by the government, such as grants and tax reductions, are significant motivators for our firm to adopt green innovation	5.203	1.440	0.924
Normative pressure (AVE = 0.783, alpha = 0.908, CR = 0.935)	1		1
Industrial associations or professional organizations encourage our firm to take energy-saving measures	5.443	1.177	0.879
The public expects our firm to adopt energy-saving measures	5.533	1.208	0.895
Customers expect to cooperate with firms with strong energy-saving awareness	5.508	1.182	0.895
The adoption of energy-saving measures in the industry is high	5.626	1.209	0.870
Mimetic pressure (AVE = 0.810, alpha = 0.883, CR = 0.927)			
Our competitors who have adopted green supply chain have greatly benefitted	5.207	1.041	0.896
Our competitors who have adopted green supply chain are favorably perceived by others in the same industry	5.252	1.064	0.907
Our competitors who have adopted green supply chain are favorably perceived by their suppliers and customers	5.224	1.076	0.897
Image management motivation (AVE = 0.834 , alpha = 0.934 , CR = 0.953)			
Our firm is concerned that other firms will think we made a poor choice	5.215	1.293	0.922
Our firm thinks other firms will think we are smart for adopting green supply chain	5.211	1.327	0.914
Our firm worries that other firms will think we are stupid for adopting green supply chain	5.293	1.302	0.908
Our firm is NOT concerned that other firms will think we are stupid for adopting green supply chain	5.297	1.315	0.910
Green supply chain adoption (AVE = 0.793 , alpha = 0.956 , CR = 0.964)			
Our firm invests resources in programs in the design for disassembly, reuse, recycling, recovery of material, components, parts	5.411	1.303	0.884
Our firm invests resources in programs in the environmentally friendly product design	5.455	1.330	0.905
Our firm invest resources in programs in the effective management of environmental risks affecting our business	5.435	1.341	0.898
Our firm invest resources in programs in the environmental improvement of packaging and transportation	5.402	1.290	0.909
Our firm invest resources in programs in the improvement of our enterprise's overall environmental situation	5.341	1.321	0.909
Our firm invest resources in programs in the environmentally friendly manufacturing processes	5.358	1.304	0.893

(Continued on following page)

TABLE 1 (Continued) Descriptive statistics and validity assessments.

Construct and indicators	Mean	STD	FL
Overall, our firm tend to invest resources in adopting green supply chain management	5.427	1.400	0.835
Environmental performance (AVE = 0.731, alpha = 0.908, CR = 0.931)			
In our firm, there is a reduction of air emission level compared it to 3 years ago	5.476	1.181	0.867
In our firm, there is a reduction of water wastage level compared it to 3 years ago	5.472	1.202	0.807
In our firm, there is a reduction of soil wastes level compared it to 3 years ago	5.492	1.281	0.853
In our firm, there is a decrease of consumption for hazardous/harmful/toxic materials compared it to 3 years ago	5.435	1.200	0.886
In our firm, there is overall improvement of an environmental situation relative to the main competitor(s) in 3 years ago	5.411	1.290	0.858
New product development performance (AVE = 0.796 , alpha = 0.949 , CR = 0.959)			
New products/services at our firm generally achieve its market share objectives	5.134	1.534	0.860
New products/services at our firm generally achieve its sales and customer use objectives	5.110	1.588	0.912
New products/services at our firm generally achieve its sales growth objectives	5.289	1.418	0.892
New products/services at our firm generally achieve its profit objectives	5.329	1.452	0.898
Our new products/services meet their performance objectives	5.297	1.456	0.907
Overall, our new products/services are successful	5.285	1.538	0.884

Note: N = 246. AVE, average variance extracted; CR, composite reliability; STD, standard deviation; FL, factor loading. Due to space constraints, detailed measurement items are omitted, which are available from the authors upon request.

TABLE 2 Correlations among	the variables	and discriminant	validity.
----------------------------	---------------	------------------	-----------

Variables		2			5	6		8	9	10	11	12
1. Firm size	-											
2. Firm age	0.298	-										
3. Industry type	-0.016	0.020	-									
4. Green process innovation	0.054	0.084	0.056	0.909								
5. Green product innovation	0.010	0.051	-0.023	0.575	0.894							
6. Coercive pressure	0.018	0.104	0.025	0.594	0.611	0.909						
7. Normative pressure	-0.040	0.028	0.008	0.562	0.636	0.558	0.885					
8. Mimetic pressure	0.168	0.097	0.094	0.561	0.583	0.513	0.565	0.900				
9. Image management motivation	0.068	0.019	-0.044	0.365	0.323	0.300	0.325	0.407	0.913			
10.Green supply chain adoption	0.025	0.059	0.010	0.633	0.702	0.688	0.634	0.597	0.268	0.891		
11. Environmental Performance	0.063	0.091	-0.013	0.502	0.620	0.522	0.565	0.466	0.252	0.644	0.855	
12. New product development performance	-0.045	0.053	0.003	0.588	0.641	0.661	0.547	0.446	0.328	0.678	0.530	0.892

Note: N = 246. Values in italicized bold denote the square root of the average variance extracted (AVE) of each construct.

4.2 Hypothesis assessment

After examining the measurement model to check the reliability and validity of the measures we used in our study, we tested the hypotheses on the internal and external forces that may drive firms to adopt a green supply chain and the conditions under which such forces more or less matter as well as how the adoption of a green supply chain may matter for firms to improve their environmental and new-product development performance. Thus, we conducted structural model analysis. Multicollinearity may emerge and become a serious concern; thus, we examined the variance inflation factor (VIF) values. The results of our VIF assessment suggested that all the VIF values were below 4.56, which is fairly below the commonly recommended cutoff of 10 (Hair et al., 1998). Therefore, multicollinearity was not likely to be a serious issue in our analysis (Burns and Bush, 2000). Nevertheless, we mean centered all the independent and moderating variables to further address the multicollinearity concerns when we developed the interaction terms for examining the moderating effect of image management motivation.



The results of the structural equation analysis are reported in Figure 2. Specifically, the coefficient of determination R^2 was 0.651, 0.494, and 0.580 for green supply chain adoption, environmental performance, and new-product development performance, high coefficient of determination R^2 respectively. The demonstrated the adequate explanatory power of our structural model analysis. Then, we examined each hypothesis by assessing its path coefficient. The path coefficient results in Figure 2 suggested a significant positive relationship between the green process innovation (b = 0.158, p < 0.05), green product innovation (b = 0.278, *p* < 0.001), coercive pressure (b = 0.283, *p* < 0.001), normative pressure (b = 0.140, p < 0.05), mimetic pressure (b = 0.122, p < 0.05), and green supply chain adoption of the firms in China's manufacturing sector. In other words, the manufacturing firms in China, which pursue green process and green product innovation and experience strong coercive, normative, and mimetic pressures, are likely to adopt a green supply chain; thus, Hypotheses one to five were supported.

Furthermore, we assessed the potential role of image management motivation in moderating the effect of green process and green product innovation and coercive, normative, and mimetic pressures on green supply chain adoption. The results in Figure 2 suggested that contrary to our expectations, image management motivation played a statistically significant but negative moderating role in the relationship between green process innovation and green supply chain adoption (b = -0.061, p < 0.01); thus, Hypothesis 6a was rejected. However, as expected, we observed the statistically significant and positive moderating effect of image management motivation on the relationship between coercive pressure and green supply chain adoption (b = 0.168, p < 0.01); thus, Hypothesis 6c was supported. Contrary to our expectations,

the results in Figure 2 show that the moderating effect of image management motivation on the relationship between green product innovation, normative pressure, mimetic pressure, and green supply chain adoption was statistically insignificant. Therefore, Hypotheses 6b, 6d, and 6e were not supported. Last, we examined the contribution of green supply chain adoption to the environmental and new-product development performance of the manufacturing firms in China. As shown in Figure 2, green supply chain adoption was statistically and positively associated with environmental (b = 0.309, p < 0.001) and new-product development (b = 0.255, p < 0.001) performance. Therefore, Hypotheses 7 and 8 were strongly supported. We discuss the potential implications of the results in the succeeding section.

5 Discussion and implications

5.1 Discussion and theoretical contributions

In this study, we attempt to determine why and the conditions under which firms may adopt green supply chain practices and the conditions under which firms can benefit from green supply chain adoption to enhance their environmental and new-product performance. We theorize and examine the central questions by developing and testing hypotheses on how internal green process and green product innovation and external institutional pressures can affect the adoption of a green supply chain. Our results demonstrate that relevant internal and external forces must be considered to understand why firms will likely adopt green supply chain practices and how the adoption of a green supply chain will benefit firms. In addition, we further theorize and examine how firms' image management motivation may moderate the effects of internal and external forces on the adoption of a green supply chain. Our study makes important contributions to the rapidly emerging literature on green supply chain management and circular economy implementation.

First, our empirical analysis reveals that green process and green product innovation have a significantly positive impact on the firms' transformation and adoption of a green supply chain, thereby extending the literature on green process and green product innovation. Green innovation can help firms gain a competitive advantage, eliminate pollutant emissions from the source in the supply chain (Chen and Liu, 2020), improve their resource and energy utilization efficiency (Severo et al., 2017; Xie et al., 2019), and coordinate the relationship between their supply chain and natural environment in a positive way and promote the transformation of their traditional supply chain into a green supply chain. In other words, firms should first improve their green product and green process innovation ability or adopt green supply chain practices before attempting to transform their traditional supply chain.

Second, coercive, normative, and mimetic pressures have a significantly positive effect on the firms' adoption of a green supply chain. Among the different types of pressures, coercive pressure from the government exerts the greatest influence on the adoption of a green supply chain. As emphasized in the literature, though firms may increase their environmental awareness in response to multiple pressures, including regulation, competition, and marketing pressures, the increased awareness will not bring about substantial changes in their green supply chain practices (Zhu et al., 2005). However, our results demonstrate that to motivate firms to adopt a green supply chain, the government may motivate them to pay considerable attention to environmental issues and force them to assume additional environmental responsibilities by implementing government policies, laws, regulations, and industry standards and raising the environmental protection awareness of various stakeholders, such as consumers. Thus, firms will have no choice but to comply with such policies to avoid administrative punishment, receive tax incentives (Nie et al., 2016; Russi et al., 2016), and gain industry access and social recognition (Zhu et al., 2013; Chu et al., 2017). When high coercive and normative pressures are imposed on firms, they will likely adopt a green supply chain.

Previous studies showed that the successful adoption of green strategies by competitors will typically exert mimetic pressure on other firms to emulate their competitors and succeed (Chu et al., 2018; Qin et al., 2021). The results of our study support this argument and suggest that the government should consider introducing a highly market-oriented competition mechanism to motivate and encourage firms to adopt a green supply chain. The government can also encourage firms to make full use of their competitive advantage to compete in a green supply chain. In addition, the government should create an excellent business environment for firms that adopt a green supply chain by actively promoting and rewarding those that are committed to utilizing advanced technologies for a green supply chain or are in the process of adopting a green supply chain.

Firms' image management motivation plays a significantly negative moderating role in the relationship between green process innovation and green supply chain adoption. In other words, the positive influence of firms' green process innovation on their green supply chain adoption will weaken as their image management motivation grows, which contradicts our hypothesis. Green process innovation generally requires the improvement of the whole process at the operational and management levels; thus, the process will consume a large amount of manpower, capital, and time, which can prevent a firm from improving its corporate image (Li et al., 2017). When individuals' image management motivation increases further, they will forgo their self-concern under pressure and choose to address the interests of others and their social environment and then engage in green behaviors (Zhang et al., 2019). However, when firms expect to improve their image within a short period of time while consuming limited resources, their excessively high image management motivation will negatively affect their green supply chain adoption and will not improve their green process innovation ability. This new finding extends the application of image management theory. Moreover, image management motivation has no significant moderating effect on the relationship between green product innovation and green supply chain adoption. Previous studies showed that the driving effect of green product innovation is influenced by internal and external factors such as technological capability, green demand, and company laws and regulations (Cai and Zhou, 2014). For firms, green product innovation can bring about new products that can realize a high resource utilization rate and low pollution emissions, which can help firms not only gain a competitive advantage but also eliminate pollutants from the source in their supply chain (Chen and Liu, 2020). Green product innovation has become increasingly common among firms attempting to adopt green supply chain practices and plays a critical role in driving such firms to adopt a green supply chain regardless of their image management motivation.

Furthermore, firms' image management motivation has no moderating effect on the relationship between either normative or mimetic pressure and their green supply chain adoption. However, image management motivation has a significantly positive moderating effect on the relationship between coercive pressure and firms' green supply chain adoption, which is consistent with the findings of prior research. In other words, as highlighted in the literature, firms may spare no expense in building their market, employer, financial, and corporate social responsibility images to leave a strong and lasting impression on their stakeholders (Highhouse et al., 2009). In terms of new-product development, to avoid violating government policies and regulations, firms will actively meet their consumers' environmental demands through new-product development and strive to create a positive image in the mind of their stakeholders (Chen, 2010). The results of our study can provide additional support for this argument by demonstrating that firms emphasize the importance of image management in their green supply chain adoption. In view of the important role of image management motivation in encouraging firms to adopt a green supply chain, the government may consider introducing incentive mechanisms that can guide or further encourage firms to adopt a green supply chain. For example, the government may identify some role model firms that have successfully adopted a green supply chain at the national, provincial, and regional levels, which would inspire other firms and innovators to adopt a green supply chain.

Last, consistent with the findings of previous studies, firms' adoption of a green supply chain plays a critical role in enhancing their environmental and new-product development performance.

Firms can adequately improve their environmental performance by adopting green supply chain practices, such as green procurement and investment recovery, and encouraging customer participation in environmental issues (Zhu et al., 2012; Wong et al., 2020). Green supply chain adoption can also improve the performance of new products and thus provide firms with strong motivation to adopt such a supply chain. Firms may also adopt green supply chain practices as an efficient mechanism to successfully develop new products.

5.2 Managerial implications

The results of our study can provide important implications for firms to effectively manage their supply chain. First, green product and green process innovation are important factors that can promote firms' adoption of a green supply chain. Thus, firms should pay attention to the importance of successful green product and process innovation in achieving successful green supply chain management. Moreover, given the important role of institutional pressures in promoting green supply chain adoption, especially as environmental protection and green development have increasingly become central national policies, firms should proactively carry out green innovation to successfully transform their traditional supply chain into a green-oriented one instead of passively reacting to the institutional pressures imposed on them. In other words, firms should be aware of the importance of actively, rather than passively, transforming environmental institutional pressures into motivation to adopt a green supply chain. In addition, image management motivation plays a negative moderating role in the relationship between green process innovation and green supply chain adoption, which may imply that firms' pursuit of short-term interests may prevent them from building a green supply chain through green process innovation. Therefore, firms should actively engage in green process innovation from a long-term perspective and establish an environmental image in the mind of the government and the public. Our results also demonstrate that the firms' green supply chain adoption positively contributes to their environmental and new-product development performance. Therefore, firms should be aware of the importance of adopting a green supply chain in enhancing their environmental and new-product development performance and make great strides in their green supply chain adoption. Specifically, firms should increase their investment and effort in environmental protection and integrate the concept of green and environmental protection into their daily business activities. Overall, the environmental and newproduct development performance of firms can help in not only their vigorous green product and green process innovation but also the resolution of national environmental problems.

The results of our study can also offer important implications for policymakers. The government should play an important role in driving firms to adopt a green supply chain, with the aim of solving environmental protection and economic development issues, by utilizing different environmental regulations, considering that coercive, normative, and mimetic institutional pressures can positively promote firms' green supply chain adoption. The government should also guide and strengthen its policy support and other institutional mechanisms to encourage firms to adopt a green supply chain. Furthermore, given the importance of green product and green process innovation in driving firms' green supply chain adoption, the government should provide effective institutional incentives and support that will encourage or help firms achieve successful green product and green process innovation. The government should also build a green supply chain ecosystem that would guide the healthy development of firms, improve its legal system, implement environmental laws and regulations that will encourage firms to assume additional corporate environmental responsibilities, and emphasize the importance of building a well-developed overall mechanism that will coordinate the environmental responsibility efforts of various firms. Image management motivation positively moderates the contribution of coercive pressure to green supply chain adoption; hence, the government should focus on monitoring and inspecting heavy-polluting firms and increase its support for green supply chain development and green technology research and development at all levels. To do so, different government levels should guide and support firms in their pollution prevention activities, organize green supply chain competitions, improve green procurement standards, and highlight the importance of green and low-carbon product procurement. In addition, the government should actively offer incentives, such as tax reductions and investment subsidies, to firms or firm projects that have achieved success in green product innovation, green process innovation, and the establishment of a green supply chain; establish a continuous monitoring and evaluation process; and contribute to the evaluation of the impact of its relevant policy regulations and measures. To help firms fully understand the importance of developing and adopting a green supply chain, the government should impose relevant institutional pressures, such as regulations, laws, policies, and industry standards. To create an environment for firms that is conducive to develop and adopt a green supply chain, the government must integrate and make full use of various institutional forces to guide and encourage society to focus on green supply chain development.

5.3 Limitations and avenues for future research

Our study has some limitations that may provide several important avenues for future research. First, our focus on Chinese firms may raise concerns about the generalizability of our findings to other research contexts. Given the heterogeneity of competitive, cultural, and institutional environments across different economies, findings based our on Chinese manufacturing firms may not apply to firms in the service industry or those in advanced developed or other emerging economies. Therefore, future research should examine the validity of our conceptual framework by collecting a complete and rich dataset from other industries or economies. Conducting comparative studies may enable scholars to further examine our conceptual framework and improve our understanding of the environment and its impact on firms' strategic green innovation or green supply chain behaviors and practices. Second, our study focuses on the factors that may drive firms' green supply chain adoption. However, the transformation of a supply chain and

10.3389/fenvs.2023.1227008

adoption of a green supply chain may involve highly complex and specific components or links within the green supply chain, such as green procurement, green manufacturing, green sales, green consumption, green recycling, and green logistics. Thus, we encourage future studies to further examine the specific components or dimensions of green supply chains and deepen our knowledge of the drivers of green supply chain adoption and its contribution to the environmental or new-product development performance of firms. Third, our study investigates only the effect of green product and green process innovation. Firms may have developed various competitive strategies to build their green innovation capabilities, and such strategies warrant further examination. Last, our research considers only two performance dimensions, namely, environmental and new-product development performance. Future research may extend our research by testing the effect of green supply chain adoption on other performance dimensions, such as cost- and flexibility-related performance outcomes, which may provide a complete picture of the impact of green supply chains. However, despite its limitations, we are confident that our study will serve as a stepping stone for the advancement of green supply chain research and stimulate future research to link green supply chain management with other innovative strategies, capabilities, and outcomes in different economies and contexts.

6 Conclusion

Owing to their increasing environmental awareness, people no longer pay attention to the innovation of simple green products and have become concerned about the entire process, from raw material procurement to sales on the market. This new trend has increasingly motivated firms to adopt a green supply chain. By integrating the ELM, institutional theory, and the image management perspective, we develop an integrative model to explore the influence of the internal and external forces driving the decision of firms to adopt a green supply chain, which in turn may positively contribute to their environmental and new-product development performance. In addition, we further examine how firms' image management motivation may moderate the effect of such forces on their green supply chain adoption. We empirically test our conceptual framework using survey data collected from a sample of 246 Chinese manufacturing firms, and the results provide broad support for our hypotheses. Our study makes

References

Amir, S., Salehi, N., Roci, M., Sweet, S., and Rashid, A. (2022). Towards circular economy: a guiding framework for circular supply chain implementation. *Bus. Strategy Environ.* 32, 2684–2701. doi:10.1002/bse.3264

Armstrong, J. S., and Overton, T. S. (1977). Estimating nonresponse bias in mail surveys. J. Mark. Res. 14 (3), 396. doi:10.2307/3150783

Arranz, C. F., Sena, V., and Kwong, C. (2022). Institutional pressures as drivers of circular economy in firms: a machine learning approach. J. Clean. Prod. 355, 131738. doi:10.1016/j.jclepro.2022.131738

Asif, M. S., Lau, H., Nakandala, D., Fan, Y., and Hurriyet, H. (2020). Adoption of green supply chain management practices through collaboration approach in developing countries – from literature review to Conceptual Framework. J. Clean. Prod. 276, 124191. doi:10.1016/j.jclepro.2020.124191

Ba, Z., Zhao, Y., Song, S., and Zhu, Q. (2022). Does the involvement of charities matter? exploring the impact of charities' reputation and social capital on medical

important contributions to the literature by expanding the scope of green supply chain research and offers useful guidelines for firms engaged in green supply chain management.

Data availability statement

The raw data supporting the conclusion of this article will be made available by the authors, without undue reservation.

Ethics statement

Ethical review and approval was not required for the study on human participants in accordance with the local legislation and institutional requirements. Written informed consent from the participants was not required to participate in this study in accordance with the national legislation and the institutional requirements.

Author contributions

All the listed authors have made substantial, direct, and intellectual contributions to the work and have approved it for publication.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

crowdfunding performance. Inf. Process. Manag. 59 (3), 102942. doi:10.1016/j.ipm. 2022.102942

Bag, S., Dhamija, P., Bryde, D. J., and Singh, R. K. (2022). Effect of eco-innovation on green supply chain management, circular economy capability, and performance of small and medium enterprises. *J. Bus. Res.* 141, 60–72. doi:10.1016/j.jbusres.2021.12.011

Benson, D. F., Brau, J. C., Cicon, J., and Ferris, S. P. (2015). Strategically camouflaged corporate governance in IPOs: entrepreneurial masking and impression management. *J. Bus. Ventur.* 30 (6), 839–864. doi:10.1016/j.jbusvent.2015.03.001

Bolino, M. C., Kacmar, K. M., Turnley, W. H., and Gilstrap, J. B. (2008). A multi-level review of impression management motives and behaviors. *J. Manage.* 34 (6), 1080–1109. doi:10.1177/0149206308324325

Bolino, M. C., and Turnley, W. H. (1999). Measuring impression management in organizations: a scale development based on the jones and pittman taxonomy. *Organ. Res. Methods* 2 (2), 187–206. doi:10.1177/109442819922005

Brislin, R. W. (1980). "Translation and content analysis of oral and written materials," in *Handbook of cross-cultural psychology*. Editors H. C. Triandis, and J. W. Berry (Boston, MA: Allyn & Bacon).

Brusset, X., and Teller, C. (2017). Supply chain capabilities, risks, and resilience. Int. J. Prod. Econ. 184, 59-68. doi:10.1016/j.ijpe.2016.09.008

Burns, A. C., and Bush, R. F. (2000). *Marketing research*. Englewood Cliffs, NJ: Prentice-Hall.

Cai, W. G., and Zhou, X. L. (2014). On the drivers of eco-innovation: empirical evidence from China. J. Clean. Prod. 79, 239-248. doi:10.1016/j.jclepro.2014.05.035

Chang, C. H. (2018). How to enhance Green Service and green product innovation performance? The roles of inward and Outward Capabilities. *Corp. Soc. Responsib. Environ. Manag.* 25 (4), 411–425. doi:10.1002/csr.1469

Chen, J., Cheng, J., and Dai, S. (2017). Regional eco-innovation in China: an analysis of eco-innovation levels and influencing factors. *J. Clean. Prod.* 153, 1–14. doi:10.1016/j. jclepro.2017.03.141

Chen, J., and Liu, L. (2020). Customer participation, and Green Product Innovation in SMEs: the mediating role of opportunity recognition and exploitation. *J. Bus. Res.* 119, 151–162. doi:10.1016/j.jbusres.2019.05.033

Chen, Y. S. (2010). The drivers of green brand equity: green brand image, green satisfaction, and green trust. J. Bus. Ethics 93 (2), 307–319. doi:10.1007/s10551-009-0223-9

Chen, Y. S., Lai, S. B., and Wen, C. T. (2006). The influence of Green Innovation performance on corporate advantage in Taiwan. *J. Bus. Ethics* 67 (4), 331–339. doi:10. 1007/s10551-006-9025-5

Cheng, J. H., and Sheu, J. B. (2012). Inter-organizational relationships and strategy quality in green supply chains—moderated by opportunistic behavior and dysfunctional conflict. *Ind. Mark. Manag.* 41 (4), 563–572. doi:10.1016/j.indmarman.2012.04.003

Cheng, P., Jiang, J., and Liu, Z. (2022). The influence of perceived external prestige on emotional labor of Frontline Employees: the mediating roles of organizational identification and impression management motive. *Int. J. Environ. Res. Publ. Health* 19 (17), 10778. doi:10.3390/ijerph191710778

Cheng, Y., Mu, D., and Zhang, Y. (2017). Mixed carbon policies based on cooperation of carbon emission reduction in supply chain. *Discrete Dyn. Nat. Soc.* 2017, 1–11. doi:10. 1155/2017/4379124

Chin, T. A., Tat, H. H., and Sulaiman, Z. (2015). Green supply chain management, environmental collaboration and sustainability performance. *Procedia CIRP* 26, 695–699. doi:10.1016/j.procir.2014.07.035

Chin, W. W. (1998). "The partial least squares approach to structural equation modeling," in *Modern methods for business research*. Editor G. Marcoulides (Mahwah, NJ: Lawrence Erlbaum Associates).

Chu, S., Yang, H., Lee, M., and Park, S. (2017). The impact of institutional pressures on green supply chain management and firm performance: top management roles and social capital. *Sustainability* 9 (5), 764. doi:10.3390/su9050764

Chu, Z., Xu, J., Lai, F., and Collins, B. J. (2018). Institutional theory and environmental pressures: the moderating effect of market uncertainty on innovation and firm performance. *IEEE Trans. Eng. Manag.* 65 (3), 392–403. doi:10.1109/tem.2018.2794453

Crano, W. D., and Prislin, R. (2006). Attitudes and persuasion. *Annu. Rev. Psychol.* 57, 345–374. doi:10.1146/annurev.psych.57.102904.190034

Dai, J., Xie, L., and Chu, Z. (2021). Developing Sustainable Supply Chain Management: the interplay of institutional pressures and sustainability capabilities. *Sustain. Prod. Consum.* 28, 254–268. doi:10.1016/j.spc.2021.04.017

Dangelico, R. M., Pujari, D., and Pontrandolfo, P. (2016). Green product innovation in manufacturing firms: a sustainability-oriented dynamic capability perspective. *Bus. Strategy Environ.* 26 (4), 490–506. doi:10.1002/bse.1932

DiMaggio, P. J., and Powell, W. W. (1983). The Iron Cage Revisited: institutional isomorphism and collective rationality in organizational fields. *Am. Sociol. Rev.* 48 (2), 147. doi:10.2307/2095101

Dubey, R., Gunasekaran, A., Childe, S. J., Papadopoulos, T., Wamba, S. F., and Song, M. (2016). Towards a theory of sustainable consumption and production: constructs and measurement. *Resour. Conserv. Recycl.* 106, 78–89. doi:10.1016/j.resconrec.2015.11.008

Fornell, C., and Cha, J. (1994). "Partial least squares," in Advanced methods of marketing research. Editor R. P. Bagozzi (Cambridge: Blackwell).

Fornell, C., and Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *J. Mark. Res.* 18 (1), 39. doi:10.2307/3151312

Geisser, S. (1975). The predictive sample reuse method with applications. J. Am. Stat. Assoc. 70 (350), 320-328. doi:10.1080/01621459.1975.10479865

Govindan, K., Kaliyan, M., Kannan, D., and Haq, A. N. (2014). Barriers analysis for green supply chain management implementation in indian industries using analytic hierarchy process. *Int. J. Prod. Econ.* 147, 555–568. doi:10.1016/j.ijpe.2013.08.018

Green, J. K., Zelbst, P. J., Meacham, J., and Bhadauria, V. S. (2012). Green supply chain management practices: impact on performance. *Supply Chain Manag.* 17 (3), 290–305. doi:10.1108/13598541211227126

Hair, J., Anderson, R., Tatham, R., and Black, W. (1998). *Multivariate data analysis*. New York: Macmillan.

Henseler, J., Ringle, C. M., and Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modeling. *J. Acad. Mark. Sci.* 43 (1), 115–135. doi:10.1007/s11747-014-0403-8

Highhouse, S., Brooks, M. E., and Gregarus, G. (2009). An organizational impression management perspective on the formation of corporate reputations. *J. Manage.* 35 (6), 1481–1493. doi:10.1177/0149206309348788

Huang, J. W., and Li, Y. H. (2015). Green Innovation and performance: the view of organizational capability and social reciprocity. *J. Bus. Ethics* 145 (2), 309–324. doi:10. 1007/s10551-015-2903-y

Huang, Y. C., and Chen, C. T. (2022). Exploring institutional pressures, firm green slack, green product innovation and green new product success: evidence from Taiwan's high-tech industries. *Technol. Forecast. Soc. Change* 174, 121196. doi:10.1016/j.techfore. 2021.121196

Hulland, J. (1999). Use of partial least squares (PLS) in strategic management research: a review of four recent studies. *Strateg. Manag. J.* 20 (2), 195–204. doi:10. 1002/(sici)1097-0266(199902)20:2<195::aid-smj13>3.0.co;2-7-7

Jeong, I., Pae, J. H., and Zhou, D. (2006). Antecedents and consequences of the strategic orientations in new product development: the case of Chinese manufacturers. *Ind. Mark. Manag.* 35 (3), 348–358. doi:10.1016/j.indmarman.2005.06.010

Kavadias, S., and Ulrich, K. T. (2020). Innovation and new product development: reflections and insights from the research published in the first 20 years of manufacturing & service operations management. *Manuf. Serv. Oper. Manag.* 22 (1), 84–92. doi:10.1287/msom.2019.0816

Khan, M., Ajmal, M. M., Jabeen, F., Talwar, S., and Dhir, A. (2023). Green supply chain management in manufacturing firms: a resource-based viewpoint. *Bus. Strategy Environ.* 32 (4), 1603–1618. doi:10.1002/bse.3207

Kitchen, J., Kerr, G., Schultz, D., McColl, R., and Pals, H. (2014). The elaboration likelihood model: review, critique and research agenda. *Eur. J. Mark.* 48 (11/12), 2033–2050. doi:10.1108/ejm-12-2011-0776

Leary, M. R., and Kowalski, R. M. (1990). Impression management: a literature review and two-component model. *Psychol. Bull.* 107 (1), 34–47. doi:10.1037/0033-2909.107.1.34

Li, D., Zheng, M., Cao, C., Chen, X., Ren, S., and Huang, M. (2017). The impact of legitimacy pressure and corporate profitability on Green Innovation: evidence from China top 100. *J. Clean. Prod.* 141, 41–49. doi:10.1016/j.jclepro.2016.08.123

Li, Y., Zhao, X., Shi, D., and Li, X. (2014). Governance of sustainable supply chains in the fast fashion industry. *Eur. Manag. J.* 32 (5), 823–836. doi:10.1016/j.emj.2014.03.001

Liang, H., Saraf, N., Hu, Q., and Xue, Y. (2007). Assimilation of enterprise systems: the effect of institutional pressures and the mediating role of Top Management. *MIS Q.* 31 (1), 59. doi:10.2307/25148781

Lin, J., Luo, Z., and Luo, X. (2020). Understanding the roles of institutional pressures and organizational innovativeness in contextualized transformation toward e-business: evidence from agricultural firms. *Int. J. Inf. Manage*. 51, 102025. doi:10.1016/j.ijinfomgt. 2019.10.010

Micheli, G. J. L., Cagno, E., Mustillo, G., and Trianni, A. (2020). Green Supply Chain Management Drivers, practices and performance: a comprehensive study on the moderators. *J. Clean. Prod.* 259, 121024. doi:10.1016/j.jclepro.2020.121024

Millson, M. R., and Wilemon, D. (2006). Driving new product success in the electrical equipment manufacturing industry. *Technovation* 26 (11), 1268–1286. doi:10.1016/j. technovation.2005.08.008

Mitra, S., and Datta, P. P. (2014). Adoption of green supply chain management practices and their impact on performance: an exploratory study of Indian manufacturing firms. *Int. J. Prod. Res.* 52 (7), 2085–2107. doi:10.1080/00207543.2013.849014

Mondal, C., and Giri, B. C. (2022a). Investigating strategies of a green closed-loop supply chain for substitutable products under government subsidy. J. Ind. Prod. Eng. 39 (4), 253–276. doi:10.1080/21681015.2021.1974962

Mondal, C., and Giri, B. C. (2022b). Retailers' competition and cooperation in a closed-loop green supply chain under governmental intervention and cap-and-trade policy. *Oper. Res.* 22, 859–894. doi:10.1007/s12351-020-00596-0

Moradi, M., and Badrinarayanan, V. (2021). The effects of brand prominence and narrative features on crowdfunding success for entrepreneurial aftermarket enterprises. *J. Bus. Res.* 124, 286–298. doi:10.1016/j.jbusres.2020.12.002

Najafi Tavani, S., Sharifi, H., and Ismail, H. (2013). A study of contingency relationships between supplier involvement, absorptive capacity and Agile Product Innovation. *Int. J. Oper. Prod. Manag.* 34 (1), 65–92. doi:10.1108/ijopm-09-2011-0331

Nakashima, K., Nose, T., and Kuriyama, S. (2006). A new approach to environmentalperformance evaluation. *Int. J. Prod. Res.* 44 (18-19), 4137–4143. doi:10.1080/ 00207540600863522

Nie, P. Y., Yang, Y. C., Chen, Y. H., and Wang, Z. H. (2016). How to subsidize energy efficiency under duopoly efficiently? *Appl. Energy* 175, 31–39. doi:10.1016/j.apenergy. 2016.04.105

Nunnally, J. C. (1978). Psychometric theory. 2nd ed. New York: McGraw-Hill.

Ogbanufe, O., Kim, D. J., and Jones, M. C. (2021). Informing cybersecurity strategic commitment through top management perceptions: the role of institutional pressures. *Inf. Manag.* 58 (7), 103507. doi:10.1016/j.im.2021.103507

Park, B. I., and Xiao, S. (2020). Is exploring dynamic capabilities important for the performance of emerging market firms? the moderating effects of entrepreneurial orientation and environmental dynamism. *Int. Stud. Manag. Organ.* 50 (1), 57–73. doi:10.1080/00208825.2019.1703378

Park, B. I., and Xiao, S. S. (2021). Doing good by combating bad in the Digital World: institutional pressures, anti-corruption practices, and competitive implications of MNE foreign subsidiaries. *J. Bus. Res.* 137, 194–205. doi:10.1016/j.jbusres.2021.08.014

Pasadeos, Y., Phelps, J., and Edison, A. (2008). Searching for our "Own theory" in advertising: an update of Research Networks. J. Mass. Commun. Q. 85 (4), 785–806. doi:10.1177/107769900808500405

Peng, M. W., and Luo, Y. (2000). Managerial ties and firm performance in a transition economy: the nature of a micro-macro link. *Acad. Manage. J.* 43 (3), 486–501. doi:10. 5465/1556406

Perks, K. J., Farache, F., Shukla, P., and Berry, A. (2013). Communicating responsibility-practicing irresponsibility in CSR advertisements. *J. Bus. Res.* 66 (10), 1881–1888. doi:10.1016/j.jbusres.2013.02.009

Petty, R. E., and Cacioppo, J. T. (1986). "The elaboration likelihood model of persuasion," in *Communication and persuasion springer series in social psychology* (New York, NY: Springer). doi:10.1007/978-1-4612-4964-1_1

Philp, M., and Nepomuceno, M. V. (2020). When the frugal become wasteful: an examination into how impression management can initiate the end-stages of consumption for frugal consumers. *Psychol. Mark.* 37 (2), 326–339. doi:10.1002/mar.21303

Pourjavad, E., and Shahin, A. (2020). A hybrid model for analyzing the risks of green supply chain in a fuzzy environment. J. Ind. Prod. Eng. 37 (8), 422–433. doi:10.1080/21681015.2020.1833995

Qin, Y., Xie, Y., and Cooke, F. L. (2021). Unethical leadership and employee knowledge-hiding behavior in the Chinese context: a moderated dual-pathway model. *Asian. Bus. Manag.* 22, 740–764. doi:10.1057/s41291-021-00154-2

Richter, N. F., Sinkovics, R. R., Ringle, C. M., and Schlägel, C. (2016). A critical look at the use of SEM in International Business Research. *Int. Mark. Rev.* 33 (3), 376–404. doi:10.1108/imr-04-2014-0148

Rostamzadeh, R., Ghorabaee, M. K., Govindan, K., Esmaeili, A., and Nobar, H. B. (2018). Evaluation of Sustainable Supply Chain Risk Management using an integrated fuzzy topsiscritic approach. J. Clean. Prod. 175, 651–669. doi:10.1016/j.jclepro.2017.12.071

Russi, D., Margue, H., Oppermann, R., and Keenleyside, C. (2016). Result-based agrienvironment measures: market-based instruments, incentives or rewards? The case of Baden-Württemberg. *Land Use Policy* 54, 69–77. doi:10.1016/j.landusepol.2016.01.012

Sarkis, J., Gonzalez-Torre, P., and Adenso-Diaz, B. (2010). Stakeholder pressure and the adoption of environmental practices: the mediating effect of training. *J. Oper. Manag.* 28 (2), 163–176. doi:10.1016/j.jom.2009.10.001

Schniederjans, D., Cao, E. S., and Schniederjans, M. (2013). Enhancing financial performance with social media: an impression management perspective. *Decis. Support Syst.* 55 (4), 911–918. doi:10.1016/j.dss.2012.12.027

Seman, N. A., Govindan, K., Mardani, A., Zakuan, N., Saman, M. Z., Hooker, R. E., et al. (2019). The mediating effect of green innovation on the relationship between green supply chain management and environmental performance. *J. Clean. Prod.* 229, 115–127. doi:10.1016/j.jclepro.2019.03.211

Severo, E. A., Guimarães, J. C., and Dorion, E. C. (2017). Cleaner production and environmental management as sustainable product innovation antecedents: a survey in Brazilian industries. *J. Clean. Prod.* 142, 87–97. doi:10.1016/j.jclepro.2016.06.090

Silva, G. M., Gomes, P. J., and Sarkis, J. (2019). The role of innovation in the implementation of green supply chain management practices. *Bus. Strategy Environ.* 28 (5), 819–832. doi:10.1002/bse.2283

Srivastava, S. K. (2007). Green supply-chain management: a state-of-the-art literature review. Int. J. Manag. Rev. 9 (1), 53–80. doi:10.1111/j.1468-2370.2007.00202.x

Stone, M. (1974). Cross-validatory choice and assessment of statistical predictions. J. R. Stat. Soc. Ser. B Methodol. 36 (2), 111–133. doi:10.1111/j.2517-6161.1974.tb00994.x

Tariq, A., Badir, Y. F., Tariq, W., and Bhutta, U. S. (2017). Drivers and consequences of green product and process innovation: a systematic review, Conceptual Framework, and Future Outlook. *Technol. Soc.* 51, 8–23. doi:10.1016/j.techsoc.2017.06.002

Tavana, M., Kian, H., Nasr, A. K., Govindan, K., and Mina, H. (2022). A comprehensive framework for sustainable closed-loop supply chain network design. *J. Clean. Prod.* 332, 129777. doi:10.1016/j.jclepro.2021.129777

Tian, H., and Wang, Z. H. (2019). Chinese Green Process Innovation in automotive painting: the strategic niche management perspective. *Int. J. Environ. Sci. Technol.* 17 (2), 993–1010. doi:10.1007/s13762-019-02530-0

van Halderen, D., Bhatt, M., Berens, G. A., Brown, J., and Van Riel, C. (2016). Managing impressions in the face of rising stakeholder pressures: examining oil companies' shifting stances in the climate change debate. *J. Bus. Ethics* 133 (3), 567–582. doi:10.1007/s10551-014-2400-8

Waltho, C., Elhedhli, S., and Gzara, F. (2019). Green Supply Chain Network Design: a review focused on policy adoption and emission quantification. *Int. J. Prod. Econ.* 208, 305–318. doi:10.1016/j.ijpe.2018.12.003

Wang, C., Zhang, Q., and Zhang, W. (2020). Corporate social responsibility, green supply chain management and firm performance: the moderating role of big-data analytics capability. *Res. Transp. Bus. Manag.* 37, 100557. doi:10.1016/j.rtbm.2020. 100557

Wang, M., Li, Y., Li, J., and Wang, Z. (2021). Green process innovation, green product innovation and its economic performance improvement paths: a survey and structural model. *J. Environ. Manage.* 297, 113282. doi:10.1016/j.jenvman.2021.113282

Wang, M., and Liu, Z. (2022). How do green innovation strategies contribute to firm performance under Supply Chain Risk? Evidence from China's manufacturing sector. *Front. Psychol.* 13, 894766. doi:10.3389/fpsyg.2022.894766

Wang, M., and Yang, W. (2021). What drives rural consumers to change e-commerce attitude and adopt e-commerce through the moderating role of corporate social responsibility in an emerging market? an empirical investigation in the Chinese context. *Sustainability* 13 (23), 13148. doi:10.3390/su132313148

Wong, C. Y., Wong, C. W., and Boon-itt, S. (2020). Effects of Green Supply Chain Integration and green innovation on environmental and cost performance. *Int. J. Prod. Res.* 58 (15), 4589–4609. doi:10.1080/00207543.2020.1756510

Wong, W. P., Sinnandavar, C. M., and Soh, K. L. (2021). The relationship between supply environment, supply chain integration and operational performance: the role of business process in curbing opportunistic behaviour. *Int. J. Prod. Econ.* 232, 107966. doi:10.1016/j.ijpe.2020.107966

Xiao, S., Lew, Y. K., and Park, B. I. (2021). International new product development performance, entrepreneurial capability, and network in high-tech ventures. *J. Bus. Res.* 124, 38–46. doi:10.1016/j.jbusres.2020.11.048

Xie, X., Huo, J., and Zou, H. (2019). Green process innovation, green product innovation, and corporate financial performance: a content analysis method. *J. Bus. Res.* 101, 697–706. doi:10.1016/j.jbusres.2019.01.010

Yang, Q., Geng, R., Jiang, Y., and Feng, T. (2021). Governance mechanisms and green customer integration in China: the joint effect of power and environmental uncertainty. *Transp. Res. E Logist. Transp. Rev.* 149, 102307. doi:10.1016/j.tre.2021.102307

Yun, S., Takeuchi, R., and Liu, W. (2007). Employee self-enhancement motives and job performance behaviors: investigating the moderating effects of employee role ambiguity and managerial perceptions of employee commitment. *J. Appl. Psychol.* 92 (3), 745–756. doi:10.1037/0021-9010.92.3.745

Zaid, A. A., Jaaron, A. A., and Bon, A. T. (2018). The impact of green human resource management and green supply chain management practices on sustainable performance: an empirical study. *J. Clean. Prod.* 204, 965–979. doi:10.1016/j.jclepro. 2018.09.062

Zameer, H., Wang, Y., Vasbieva, D. G., and Abbas, Q. (2021). Exploring a pathway to carbon neutrality via reinforcing environmental performance through green process innovation, environmental orientation and green competitive advantage. *J. Environ. Manage.* 296, 113383. doi:10.1016/j.jenvman.2021.113383

Zhang, Y., Ao, J., and Deng, J. (2019). The influence of high-low power on green consumption: the moderating effect of impression management motivation. *Sustainability* 11 (16), 4287. doi:10.3390/su11164287

Zhang, Y., Yang, J., and Liu, M. (2022). Enterprises' energy-saving capability: empirical study from a dynamic capability perspective. *Renew. Sustain. Energy Rev.* 162, 112450. doi:10.1016/j.rser.2022.112450

Zhu, Q., and Sarkis, J. (2007). The moderating effects of institutional pressures on emergent green supply chain practices and performance. *Int. J. Prod. Res.* 45 (18-19), 4333–4355. doi:10.1080/00207540701440345

Zhu, Q., Sarkis, J., and Geng, Y. (2005). Green supply chain management in China: pressures, practices and performance. *Int. J. Oper. Prod. Manag.* 25 (5), 449–468. doi:10. 1108/01443570510593148

Zhu, Q., Sarkis, J., and Lai, K. H. (2012). Examining the effects of green supply chain management practices and their mediations on performance improvements. *Int. J. Prod. Res.* 50 (5), 1377–1394. doi:10.1080/00207543.2011.571937

Zhu, Q., Sarkis, J., and Lai, K. H. (2013). Institutional-based antecedents and performance outcomes of internal and external green supply chain management practices. *J. Purch. Supply Manag.* 19 (2), 106–117. doi:10.1016/j.pursup.2012.12.001